Author's response to reviews

Title: Validity of self-reported leisure-time sedentary behavior in adolescents

Authors:

- Olivia Affuso (oaffuso@uab.edu)
- June Stevens (june_stevens@unc.edu)
- Diane Catellier (diane_catellier@unc.edu)
- Robert McMurray (exphys@email.unc.edu)
- Dianne Ward (dward@email.unc.edu)
- Leslie Lytle (lytle@epi.umn.edu)
- Melinda Sothern (mosthe@lsuhsc.edu)
- Deborah Young (dryoung@umd.edu)

Version: 4 Date: 10 February 2011

Author's response to reviews:

I have made the required changes to the document including updates to the references and author names.